Author's response to reviews

Title: Increased Risk of Cancer among Relatives of Patients with Lung Cancer in China

Authors:

Yongtang Jin (jinedu@163.com)
Yingchun Xu (ycxu66@163.com)
Ming Xu (mingxu@ah172.com.cn)
Saoli Xue (xuesl@sina.com)

Version: 12 Date: 13 October 2005

Author's response to reviews: see over
Our changes point-by-point as follow:

(1) We removed the asterisk from the end of the title;

(2) Yongtang Jin" was changed to Yongtang Jin*;

(3) P.R.China was changed to P R China;

(4) We move “Abbreviations” to before the authors’ contributions section;

(5) We made the TABLE 4 simple and all tables were incorporated into the main manuscript after the references;

(6) All tables were oriented in portrait.